import { TypeList } from "docs-ui"

# promotion

## Interfaces

- [AddItemAdjustmentAction](../../promotion/interfaces/promotion.AddItemAdjustmentAction/page.mdx)
- [AddPromotionsToCampaignDTO](../../promotion/interfaces/promotion.AddPromotionsToCampaignDTO/page.mdx)
- [AddShippingMethodAdjustment](../../promotion/interfaces/promotion.AddShippingMethodAdjustment/page.mdx)
- [ApplicationMethodDTO](../../promotion/interfaces/promotion.ApplicationMethodDTO/page.mdx)
- [BaseFilterable](../../promotion/interfaces/promotion.BaseFilterable/page.mdx)
- [CampaignBudgetDTO](../../promotion/interfaces/promotion.CampaignBudgetDTO/page.mdx)
- [CampaignBudgetExceededAction](../../promotion/interfaces/promotion.CampaignBudgetExceededAction/page.mdx)
- [CampaignDTO](../../promotion/interfaces/promotion.CampaignDTO/page.mdx)
- [ComputeActionAdjustmentLine](../../promotion/interfaces/promotion.ComputeActionAdjustmentLine/page.mdx)
- [ComputeActionContext](../../promotion/interfaces/promotion.ComputeActionContext/page.mdx)
- [ComputeActionItemLine](../../promotion/interfaces/promotion.ComputeActionItemLine/page.mdx)
- [ComputeActionShippingLine](../../promotion/interfaces/promotion.ComputeActionShippingLine/page.mdx)
- [Context](../../promotion/interfaces/promotion.Context/page.mdx)
- [CreateApplicationMethodDTO](../../promotion/interfaces/promotion.CreateApplicationMethodDTO/page.mdx)
- [CreateCampaignBudgetDTO](../../promotion/interfaces/promotion.CreateCampaignBudgetDTO/page.mdx)
- [CreateCampaignDTO](../../promotion/interfaces/promotion.CreateCampaignDTO/page.mdx)
- [CreatePromotionDTO](../../promotion/interfaces/promotion.CreatePromotionDTO/page.mdx)
- [CreatePromotionRuleDTO](../../promotion/interfaces/promotion.CreatePromotionRuleDTO/page.mdx)
- [FilterableCampaignProps](../../promotion/interfaces/promotion.FilterableCampaignProps/page.mdx)
- [FilterablePromotionProps](../../promotion/interfaces/promotion.FilterablePromotionProps/page.mdx)
- [FilterablePromotionRuleProps](../../promotion/interfaces/promotion.FilterablePromotionRuleProps/page.mdx)
- [FindConfig](../../promotion/interfaces/promotion.FindConfig/page.mdx)
- [IBigNumber](../../promotion/interfaces/promotion.IBigNumber/page.mdx)
- [IMessageAggregator](../../promotion/interfaces/promotion.IMessageAggregator/page.mdx)
- [IModuleService](../../promotion/interfaces/promotion.IModuleService/page.mdx)
- [IPromotionModuleService](../../promotion/interfaces/promotion.IPromotionModuleService/page.mdx)
- [JoinerServiceConfig](../../promotion/interfaces/promotion.JoinerServiceConfig/page.mdx)
- [JoinerServiceConfigAlias](../../promotion/interfaces/promotion.JoinerServiceConfigAlias/page.mdx)
- [MessageAggregatorFormat](../../promotion/interfaces/promotion.MessageAggregatorFormat/page.mdx)
- [PromotionDTO](../../promotion/interfaces/promotion.PromotionDTO/page.mdx)
- [PromotionRuleDTO](../../promotion/interfaces/promotion.PromotionRuleDTO/page.mdx)
- [PromotionRuleValueDTO](../../promotion/interfaces/promotion.PromotionRuleValueDTO/page.mdx)
- [RemoveItemAdjustmentAction](../../promotion/interfaces/promotion.RemoveItemAdjustmentAction/page.mdx)
- [RemovePromotionsFromCampaignDTO](../../promotion/interfaces/promotion.RemovePromotionsFromCampaignDTO/page.mdx)
- [RemoveShippingMethodAdjustment](../../promotion/interfaces/promotion.RemoveShippingMethodAdjustment/page.mdx)
- [RestoreReturn](../../promotion/interfaces/promotion.RestoreReturn/page.mdx)
- [SoftDeleteReturn](../../promotion/interfaces/promotion.SoftDeleteReturn/page.mdx)
- [UpdateApplicationMethodDTO](../../promotion/interfaces/promotion.UpdateApplicationMethodDTO/page.mdx)
- [UpdateCampaignBudgetDTO](../../promotion/interfaces/promotion.UpdateCampaignBudgetDTO/page.mdx)
- [UpdateCampaignDTO](../../promotion/interfaces/promotion.UpdateCampaignDTO/page.mdx)
- [UpdatePromotionDTO](../../promotion/interfaces/promotion.UpdatePromotionDTO/page.mdx)
- [UpdatePromotionRuleDTO](../../promotion/interfaces/promotion.UpdatePromotionRuleDTO/page.mdx)

## Type Aliases

- [ApplicationMethodAllocationValues](../../promotion/types/promotion.ApplicationMethodAllocationValues/page.mdx)
- [ApplicationMethodTargetTypeValues](../../promotion/types/promotion.ApplicationMethodTargetTypeValues/page.mdx)
- [ApplicationMethodTypeValues](../../promotion/types/promotion.ApplicationMethodTypeValues/page.mdx)
- [BigNumberInput](../../promotion/types/promotion.BigNumberInput/page.mdx)
- [BigNumberRawValue](../../promotion/types/promotion.BigNumberRawValue/page.mdx)
- [CampaignBudgetTypeValues](../../promotion/types/promotion.CampaignBudgetTypeValues/page.mdx)
- [ComputeActions](../../promotion/types/promotion.ComputeActions/page.mdx)
- [Constructor](../../promotion/types/promotion.Constructor/page.mdx)
- [Event](../../promotion/types/promotion.Event/page.mdx)
- [EventMetadata](../../promotion/types/promotion.EventMetadata/page.mdx)
- [ExpandScalar](../../promotion/types/promotion.ExpandScalar/page.mdx)
- [FilterQuery](../../promotion/types/promotion.FilterQuery/page.mdx)
- [FilterQueryProperties](../../promotion/types/promotion.FilterQueryProperties/page.mdx)
- [FilterValue](../../promotion/types/promotion.FilterValue/page.mdx)
- [FilterValue2](../../promotion/types/promotion.FilterValue2/page.mdx)
- [FindConfigOrder](../../promotion/types/promotion.FindConfigOrder/page.mdx)
- [JoinerRelationship](../../promotion/types/promotion.JoinerRelationship/page.mdx)
- [LinkModulesExtraFields](../../promotion/types/promotion.LinkModulesExtraFields/page.mdx)
- [Message](../../promotion/types/promotion.Message/page.mdx)
- [ModuleJoinerConfig](../../promotion/types/promotion.ModuleJoinerConfig/page.mdx)
- [ModuleJoinerRelationship](../../promotion/types/promotion.ModuleJoinerRelationship/page.mdx)
- [OperatorMap](../../promotion/types/promotion.OperatorMap/page.mdx)
- [PrevLimit](../../promotion/types/promotion.PrevLimit/page.mdx)
- [Primary](../../promotion/types/promotion.Primary/page.mdx)
- [PromotionRuleOperatorValues](../../promotion/types/promotion.PromotionRuleOperatorValues/page.mdx)
- [PromotionStatusValues](../../promotion/types/promotion.PromotionStatusValues/page.mdx)
- [PromotionTypeValues](../../promotion/types/promotion.PromotionTypeValues/page.mdx)
- [Query](../../promotion/types/promotion.Query/page.mdx)
- [RawMessageFormat](../../promotion/types/promotion.RawMessageFormat/page.mdx)
- [ReadonlyPrimary](../../promotion/types/promotion.ReadonlyPrimary/page.mdx)
- [Scalar](../../promotion/types/promotion.Scalar/page.mdx)
- [UsageComputedActions](../../promotion/types/promotion.UsageComputedActions/page.mdx)
